Author's response to reviews

Title: A phase II, Randomized, Sham-Controlled, Double-Blinded Trial Testing the Safety and Efficacy of the Coronary Sinus Reducer in Patients with Refractory Angina: Rationale and Design.

Authors:

Marc E Jolicoeur (marc.jolicoeur@icm-mhi.org)
Shmuel Banai (banais@netvision.net.il)
Timothy D Henry (henry003@umn.edu)
Marc Schwartz (mschwartz@neovasc.com)
Serge Doucet (serge.doucet@icm-mhi.org)
Christopher J. White (drcjwhite@gmail.com)
Elazer Edelman (ere@mit.edu)
Stefan Verheye Verheye (stefan.verheye@telenet.be)

Version: 2 Date: 16 August 2012

Author’s response to reviews:

As requested, we have completed the formatting changes

We also confirm that the documentation refers to the correct study and we can tell which body granted the approvals and that the full ethical approval has been given (i.e. not with conditions)